Author's response to reviews

Title: War-related psychological sequelae among emergency department patients in the former Republic of Yugoslavia

Authors:

Brett D Nelson BA (bnelso1@jhmi.edu)
Dr William G Fernandez MD, MPH (william.fernandez@bmc.org)
Dr Sandro Galea MD, DrPH (sgalea@nyam.org)
Sarah Sisco MPH, MSSW (ssisco@nyam.org)
Kerry Dierberg BS (kdierbe1@jhmi.edu)
Gordana SUBARIC Gorgieva MD, PhD (subariogorgieva@yahoo.com)
Arijit K Nandi BS (akn2005@columbia.edu)
Jennifer Ahern MPH (jahern@nyam.org)
Dr Mihajlo Mitrovic MD (gvozden@eunet.yu)
Michael VanRooyen MD, MPH (mvanrooy@jhmi.edu)
David Vlahov PhD (dvlahov@nyam.org)

Version: 6 Date: 25 May 2004

Iratxe Puebla
Assistant Editor
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Re: MS 7642205492459296

Dear Ms Puebla,

I am submitting the revisions to the copy-edited manuscript, "War-related psychological sequelae among emergency department patients in the former Republic of Yugoslavia".

We have made the updates as per your editorial suggestions (see below):

Title page: complete addresses (and email addresses) for all authors. Please note: No "official" postal code exists for the Laplje Selo Hospital, as there is no formal mail delivery in that area of Kosovo.

Abstract/Methods (and Introduction, 3rd paragraph): References to post-traumatic stress (PTSD) have been changed to 'post-traumatic stress disorder'

Methods: Reference to 'ethanol' have been changed to 'alcohol'

Results: Tables 1 and 2 have been changed to reflect the analysis for living with a spouse or partner.

Authors' contribution: David Vlahov's contribution to the manuscript has been updated.

In terms of track changes, we have accepted all editing changes that have been offered by the
editorial staff, and have been saved in a new file. Consequently, a few minor wording changes in have been made to clarify the syntax of the final copyedited paper.

My colleagues and I appreciate the opportunity to submit to your journal, and look forward to working with you again in the future.

Regards,

Bill Fernandez
Assistant Professor of Emergency Medicine
Department of Emergency Medicine
Boston University School of Medicine
Boston Medical Center
Office: 617.414.4927 / Fax: 617.414.7759
Email: William.Fernandez@bmc.org